Author's response to reviews

Title: A Pitfall in the Interpretation of Plain Abdominal Radiograph in Neonates: a case presentation.

Authors:

  Martin J Gillies (martingillies@gmail.com)
  Moti M Chowdhury (moti3773@yahoo.ie)
  Kokila Lakhoo (kokila.lakhoo@paediatrics.ox.ac.uk)

Version: 2  Date: 26 March 2008

Author's response to reviews:

The following revisions were made, on the emailed instructions sent by the editor of the journal. These revisions are stylistic, rather than to content:

1. In the title 'case study' was changed to 'case presentation'.

2. The abstract has been reformulated to include the headings introduction, case presentation and conclusion. I have altered the sentence structure of some of the abstract sentences to attempt to make a more direct writing style.

I have made minor alterations to the introduction too, in an attempt to make it read more fluently. This has involved alterations to sentence structure.

best wishes

Martin Gillies